

# GOPEN ACCESS

**Citation:** Besenbacher S, Sulem P, Helgason A, Helgason H, Kristjansson H, Jonasdottir A, et al. (2016) Multi-nucleotide *de novo* Mutations in Humans. PLoS Genet 12(11): e1006315. doi:10.1371/journal.pgen.1006315

Editor: Dmitri A. Petrov, Stanford University, UNITED STATES

Received: February 2, 2016

Accepted: August 22, 2016

Published: November 15, 2016

**Copyright:** © 2016 Besenbacher et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** The chromosomes, positions and alleles of all the found de novo mutations can be found in <u>S1 Table</u>. Information on all the analysed clusters of de novo mutations can be found in <u>S2 Table</u>.

**Funding:** SB's salary was paid by a "Young Investigator" grant (VKR023129) from the Villum Foundation (<u>http://veluxfoundations.dk/en</u>). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

# Multi-nucleotide *de novo* Mutations in Humans

Søren Besenbacher<sup>1</sup>\*, Patrick Sulem<sup>2</sup>, Agnar Helgason<sup>2,3</sup>, Hannes Helgason<sup>2,4</sup>, Helgi Kristjansson<sup>2</sup>, Aslaug Jonasdottir<sup>2</sup>, Adalbjorg Jonasdottir<sup>2</sup>, Olafur Th. Magnusson<sup>2</sup>, Unnur Thorsteinsdottir<sup>2,5</sup>, Gisli Masson<sup>2</sup>, Augustine Kong<sup>2</sup>, Daniel F. Gudbjartsson<sup>2,4</sup>, Kari Stefansson<sup>2,5</sup>

1 Department of Molecular Medicine, Aarhus University, Denmark, 2 deCODE genetics/Amgen, Inc., Iceland, 3 Department of Anthropology, University of Iceland, Iceland, 4 School of Engineering and Natural Sciences, University of Iceland, Iceland, 5 Faculty of Medicine, University of Iceland, Iceland

\* besenbacher@clin.au.dk

# Abstract

Mutation of the DNA molecule is one of the most fundamental processes in biology. In this study, we use 283 parent-offspring trios to estimate the rate of mutation for both single nucleotide variants (SNVs) and short length variants (indels) in humans and examine the mutation process. We found 17812 SNVs, corresponding to a mutation rate of  $1.29 \times 10^{-8}$  per position per generation (PPPG) and 1282 indels corresponding to a rate of  $9.29 \times 10^{-10}$  PPPG. We estimate that around 3% of human *de novo* SNVs are part of a multi-nucleotide mutation (MNM), with 558 (3.1%) of mutations positioned less than 20kb from another mutation in the same individual (median distance of 525bp). The rate of *de novo* mutations is greater in late replicating regions (p =  $8.29 \times 10^{-19}$ ) and nearer recombination events (p = 0.0038) than elsewhere in the genome.

# Author Summary

In each generation new genetic variants are introduced by mutations. In this study we use whole genome sequence data from Icelandic families to directly observe such new mutations. Our estimate of the mutation rate implies that a newborn with 30-year-old parents will on average carry 75 new SNV mutations and 6 new short indel mutations. We observe that some of the found mutations occur much closer together than would be expected by chance. Our analysis shows that mutational hotspots cannot explain this clustering, instead the clustering mutations are likely created by a single mutational event. We observe a different composition for mutations that cluster very close together compared to more distant clustering mutations. This suggests that there is likely more than one type of underlying mutational mechanism creating the multi nucleotide mutation events. We furthermore observe a higher mutation rate near recombination events but find that this effect cannot explain the large number of clustering mutations that we observe. PLOS GENETICS

**Competing Interests:** The authors have declared that no competing interests exist.

#### Introduction

Germline *de novo* mutations–i.e. mutations that occur during the formation of egg and sperm cells–are ultimately responsible for all heritable traits and evolutionary adaptations. Knowl-edge about the properties of these mutations is important for timing events in evolutionary history and understanding the causes of phenotypic diversity, such as disease. It is now possible to directly and comprehensively identify *de novo* mutations using whole genome sequence (WGS) data from nuclear families. Thus, we are now better equipped to answer questions such as: What is the mutation rate in humans? What affects the differences between the numbers and types of mutations within and between individuals? And are all mutations the result of independent events?

Several recent studies have estimated the mutation rate using WGS data from nuclear families [1-5]. However, calculating a rate estimate is not a trivial task and some uncertainty about the actual rate remains [6]. Comparisons between individuals show that paternal age explains most of the diversity in the number of *de novo* single nucleotide variants (SNVs), but it is less clear whether there are contributions from other factors such as maternal age or environmental exposure. Because *de novo* indels are somewhat rarer than *de novo* SNVs, and harder to detect, most studies have excluded them and less is therefore known about their mutation rate and the impact of factors such as parental age.

The germline mutation rate varies across the human genome at both fine and broad scales [6]. The most important factors affecting variation in the fine scale mutation rate are sequence context and methylation. At a broader scale, replication timing has been shown to be influential with significantly more mutations occurring in late-replicating regions [7]. The rate of recombination is positively correlated with genetic diversity in the human genome, but there has been much debate over whether this is due to increased efficiency of selection, confounding factors or a mutagenic effect of recombination [8–10].

It is often assumed that all mutations are the result of independents event even though several lines of evidence call this assumption into question [6]. Both family studies of germline mutations [3,11] and population data [12] have shown that mutations close to each other occur much more frequently than would be expected by chance, presumably due to single events that give rise to multiple nucleotide mutations (MNMs). In this paper we try to characterize such MNM events and estimate how common they are, which is important in order to understand their impact on human evolution and health. In addition we examine the potential mechanisms that could be responsible for different kinds of MNM events.

# Results

#### The rate of de novo mutations

As a part of a large sequencing project in Iceland [13], we have obtained whole genome sequences (WGS) of 283 parent-offspring trios. Variant calling was performed using GATK and conservative filtering criteria were applied to identify a high-confidence set of autosomal *de novo* mutations (see methods). This resulted in a set of 17812 SNVs (avg. of 63 per individual), corresponding to a mutation rate of  $1.29 \times 10^{-8}$  per position per generation (PPPG), with a 95% confidence interval (c.i.) of  $1.27 \times 10^{-8}$  to  $1.30 \times 10^{-8}$ . We also identified 929 short deletions and 353 short insertions (less than 35bp), corresponding to a combined indel mutation rate of  $9.29 \times 10^{-10}$  PPPG (c.i.:  $8.79 \times 10^{-10} - 9.82 \times 10^{-10}$ ). This indel mutation rate falls between two recent estimates based on whole genome sequencing of trios [1,14]. The ratio between the *de novo* SNV rate and indel rate (13.78) is the same as that observed between the overall number of segregating SNVs and short indels in the Icelandic population (13.68)[13].

. . . . . . . . . . . .

The denominators of the rate estimates were calculated separately for each individual and independently for SNVs and indels using a probabilistic method (see <u>methods</u>). Using this probabilistic method, we estimate that we can correctly identify 91.3% of the autosomal *de novo* SNVs and 90.9% of short *de novo* indels.

For both SNVs and indels, the rate of *de novo* events per offspring is positively correlated with the age of parents (see Fig 1). The ages of the parents are, however, highly correlated ( $r^2 = 0.65$ ). A multiple regression analysis shows that while the correlation with the mutation rate is primarily driven by father's age at conception (amounting to roughly 1.7 autosomal mutations for each additional year,  $p = 8.3 \times 10^{-26}$ ), there is also a borderline effect of mother's age (~0.34 mutations per year, p = 0.041). Similarly, indels are primarily driven by father's age (~0.1 per year,  $6.1 \times 10^{-3}$ ), with a non-significant effect of mother's age (~0.01 per year, p = 0.75). The average age of the fathers and mothers is 31.6 and 28.9 respectively. The finding of a maternal age effect for SNV mutations is consistent with a recent study that found a maternal age effect of 0.35 mutations per year, but our estimate of the paternal age effect is significantly higher than what was reported in that study (0.64 additional mutations per year)[15].

As the number of mutations per generation is strongly affected by the length of the generation interval, it is more informative to report rates in the scale of years. We estimate the SNV mutation rate to be  $4.25 \times 10^{-10}$  (95% c.i.:  $4.18 \times 10^{-10}$ – $4.31 \times 10^{-10}$ ) PPPY. This is slightly higher than previous estimates based on parent-offspring trios, but is nonetheless approximately a factor of 2 lower than the rate typically used to estimate speciation times between



Fig 1. The correlation between the estimated mutation rate for each child and the age of the parents.

doi:10.1371/journal.pgen.1006315.g001

humans and closely related species [<u>16</u>]. <u>Table 1</u> shows mutation rates per year for different types of mutations.

The estimated indel rate is  $3.07 \times 10^{-11}$  PPPY (95% c.i.:  $2.91 \times 10^{-11}$ – $3.25 \times 10^{-11}$ ). The observed deletion to insertion ratio of 2.6 is consistent with a ratio of 2.8 observed for exonic *de novo* indels [17], but it is significantly higher than a ratio of ~2 estimated by using related species to determine the ancestral states of segregating human indels [18]. The observed distribution of indel lengths shows that the two most common types of *de novo* indels are 1 bp deletions (33.8%) and 1 bp insertions (19.7%) (see S1 Fig). The largest deletion that we observe is 35 bp long, while the largest insertion is only 8 bp long. The absence of larger insertions is probably because those variants are difficult to call using short read and therefore the true deletion to insertion ratio is likely to be smaller than our estimate.

#### Multi-nucleotide mutation events

Mutations are typically assumed to occur uniquely and independently in the genome, even though several lines of evidence call these assumptions into question [6]. Testing for the random distribution of distances between neighboring *de novo* SNVs in our data, we see a significant overrepresentation of mutations close to each other. Fig 2A shows the distribution of observed distances compared to the null expectation that mutations occur independently. These results show that the clustering of mutations occurs solely within individuals and that the distances between mutations from different individuals are consistent with the null expectation. Thus, there is no sign of mutational sites shared by many individuals.

Clustering of mutations within individuals has been reported in previous studies and can be explained by MNM events [3,19-21]. One study estimated that 1.9% of all mutations are part of a MNM event with positions separated by less than 20bp [22]. Recent studies of *de novo* mutations found enrichment of mutations that are several kb apart [3,21] and it is possible that these more distant clustered mutations are also the result of a single mutational event. We find that 2.4% (435 of 17812) of *de novo* SNVs are accompanied by another mutation in the same individual less than 5kb away. Assuming that mutations are independent we would expect 0.03% (95% c.i.: 0.00%–0.07%) of mutations within this distance of each other. Using a less conservative criterion, we find that 3.1% (558 of 17812) of the *de novo* SNVs are accompanied by another mutation less than 20kb away compared to an expectation of 0.11% (95% c.i.:

Mutation Type	Sequence context	Transition vs. Transversion	Number of mutations	Mutation rate PPPY x 10 <sup>10</sup> (95% c.i.)	
SNV	CpG	Transition	2984	39.81	(38.40–41.26)
		Transversion	281	3.75	(3.34–4.21)
		All	3265	43.55	(42.09–45.07)
	nonCpG Strong (C or G)	Transition	4264	2.64	(2.56–2.72)
		Transversion	3019	1.87	(1.80–1.93)
		All	7283	4.50	(4.40–4.61)
	Weak (A or T)	Transition	4758	1.90	(1.85–1.96)
		Transversion	2506	1.00	(0.96–1.04)
		All	7264	2.91	(2.84–2.97)
Insertion	All		353	0.08	(0.08–0.09)
Deletion	All		929	0.22	(0.21–0.24)

Table 1. The rate of different types of mutations.

The rates per position per year (PPPY) for different types of mutations. G and C base pairs are referred to as strong because they are bound by three hydrogen bonds while weak (A and T) base pairs are bound by two hydrogen bonds.

doi:10.1371/journal.pgen.1006315.t001



**Fig 2. Clustering mutations.** (a) The red line shows a QQ-plot of the observed distances between all pairs of mutations (both within and between individuals) compared to the expected distances assuming independence. The green line shows a QQ plot based only on distances between mutations that occurred in the same individual. The blue line shows a QQ plot based only on distances between mutations that occurred in the number of mutations per cluster. (c) Histogram showing the distribution of distances to the nearest mutation in the same individual.

doi:10.1371/journal.pgen.1006315.g002

0.03%–0.20%). We therefore conclude that around 3% of SNVs are likely the result of MNM events. This estimate is significantly larger ( $p = 3.5 \times 10^{-19}$ ) than what was reported in a recent article [11] where 1.46% (161 of 11020) of mutations clustered within 20kb. This discrepancy cannot be fully explained by the fact that our study has higher sequence coverage and that our mutation calls thus have higher specificity. If we subsample our mutations so that we have the same average number of mutations per individual as in [11] (44.1 compared to our 62.9) we still observe that 2.4% (95% c.i.: 2.1%–2.6%) of mutations cluster within 20kb.

The 558 SNV mutations can be grouped into 247 MNM clusters, most of them with just two mutations, but with the largest cluster containing 8 mutations (see Fig 2C). The majority (315 of 558) of clustered mutations are less than 2kb from another mutation (median distance = 525 bp) and 17% (108) are immediately adjoining. Considering these 108 adjoining mutations as 54 tandem mutation events we estimate that the tandem mutation rate is 0.30% (95% c.i.: 0.23%–0.40%) of the single nucleotide mutation rate. This estimate is not far from the estimate of 0.4% that was recently calculated in a meta-analysis of 7 different studies that had estimated the tandem mutation rate [23].

We validated a subset of the 558 clustered mutations using Sanger sequencing. For 11 of the 57 pairs of clustered mutations that we tried to validate, sequencing failed in at least one trio member for at least one of the variants. The remaining 46 pairs of variants were all validated as genuine *de novo* variants that were present in the child, but not in the parents, yielding an estimated false discovery rate of 0.0% (95% c.i.: 0.0%–4.0%). The pairs were selected to validate both clusters spanning few bases and those spanning several kilobases, such that 19 were separated by <100bp, 19 between 100–2000 bp and the remaining 19 pairs between 2kb and 20kb.

We expect all the mutations in a MNM cluster to originate in the same parent and the alternative alleles should thus be found on the same chromosome. This can be tested when two mutations are close enough to be found on the same read or on paired-end reads from the same template molecule. We observed 159 pairs of positions with mutations (121 clusters) that were covered by single or paired-end reads. For 158 of these, the alternative alleles were found on the same template molecule (i.e. chromosome). For one pair of mutations, the information was contradictory with 4 reads supporting one phasing and 11 reads supporting a different phasing. The fact that the clustering mutations occur on the same chromosome supports our conclusion that they are created by a single mutational event.

If we consider both SNVs and indels together, 656 mutations (in 286 MNM clusters) are accompanied by another mutation in the same individual less than 20kb away. We observe 52 clusters that contain indels and 10 of these clusters consist solely of indels and the remaining 42 contain both SNVs and indels.

#### Mutational mechanisms

We next examined more closely the mutations that cluster within individuals to gain insight into the mutational mechanisms that cause MNM events. Fig 3 shows the relative rates of all six possible types of SNV mutations, ignoring strand differences (i.e.  $C \rightarrow T$  is the same as  $G \rightarrow A$ ). The results show that the frequencies of mutation types in clusters vary as a function of the distance between the mutations. Mutations less than 10bp from each other show a significant overrepresentation of  $A \rightarrow T$  and the tandem mutations also show a significant overrepresentation of  $C \rightarrow A$  mutations. Such a pattern is not observed in the 10bp-20kb categories, which instead show an overrepresentation of  $C \rightarrow G$  mutations. The mutational spectrums for each of the different groups of clustering mutations shown in Fig 3 are all significantly different compared to the non-clustering mutations (the right most bin in Fig 3)(all comparisons have a pvalue less than  $3 \times 10^{-3}$ ). Furthermore the tandem mutations are significantly different from all the other groups of clustering mutations and so is the group of mutations that cluster 2 to 10bp apart. These results suggest that different mutational mechanisms may underlie the tandem mutations, the mutations that are 2 to 10bp apart and the more distal clustered mutations.

**Possible mechanisms creating tandem mutations.** A recent paper described the use of population data to infer past MNM events in humans and reported a skewed distribution of ancestral and derived sites in tandem mutations [24]. They reported that  $GC \rightarrow AA$  and  $GA \rightarrow TT$  and their reverse complements were overrepresented and accounted for 27% of the tandem mutations. In our data, these two types of mutations make up 20% (11/54) of the



Distance to hearest mutation in same individual (op)

Fig 3. Distribution of mutation types. The relative distribution of different types of mutations stratified by the distance to nearest mutation in the same individual. The error bars are 95% confidence intervals.

doi:10.1371/journal.pgen.1006315.g003

tandem mutation pairs, which is much more than the 2.2% that would be expected if the mutations were independent ( $p = 2.3 \times 10^{-8}$ ). These two types of mutations have previously been shown to dominate the spectrum of mutations that are introduced by the error-prone polymerase  $\zeta$  [25], suggesting that low-fidelity replication by polymerase  $\zeta$  caused at least some of the tandem mutations that we observe. Experiments in yeast have also shown that polymerase  $\zeta$  can produce complex mutations that include both tandem mutation and indels [25,26]. We observe that 13%(7/54) of the tandem mutations are accompanied with a *de novo* indel less than 10 bp away.

**Possible mechanisms causing mutations occurring 10bp-20kb apart.** The overrepresentation of  $C \rightarrow G$  among clustering mutations was also noticed in [11], but the mechanism responsible for these remains unknown. The error prone polymerase REV1 is known to cause  $C \rightarrow G$  mutations at abasic sites created by cytosine deamination and may thus be involved in the creation of these mutations [27,28]. The activity of the APOBEC cytosine deaminases is known to cause clusters of  $C \rightarrow G$  and  $C \rightarrow T$  mutations in several types of cancers by deaminating cytosines in stretches of single stranded DNA [29]. The deamination induced by APOBEC occurs primarily at TCW sites (the underlined base is the mutated base, and W means A or T). If we look at the mutations that cluster 10bp to 20kb apart we do not observe that a significantly larger fraction of the C $\rightarrow$ G mutations in this group occur at TCW sites (23.7%) compared to the non-clustering  $C \rightarrow G$  mutations (22.2%) (p = 0.71). Nor do we observe a significantly increased fraction of clustering C $\rightarrow$ T mutations at TCW sites (p = 0.15). This means that we see no evidence of APOBEC activity being the cause of the cytosine deamination leading up to the cytosine mutations. Instead we observe a more general pattern with enrichment at cytosine positions that are not preceded by another cytosine or followed by a guanine base (i.e. DCH $\rightarrow$ G where D = A/G/T and H = A/C/T). 85.1% of the clustering C $\rightarrow$ G mutations match this DCH $\rightarrow$ G motif compared to only 68.3% of the non-clustering C $\rightarrow$ G mutations (p =  $2.0 \times 10^{-4}$ ). The results in Fig 4 also show that there are relatively fewer CpG transitions among the clustering mutations than the non-clustering. The percentage of  $C \rightarrow T$ transitions that occur at CpG-sites is 41.8% (95% c.i. 40.6%-42.9%) for non-clustered mutations, but only 14.7% (95% c.i. 10.0%-21.2%) for clustered mutations. This is evidence that





doi:10.1371/journal.pgen.1006315.g004

cytosine methylation is not responsible for the clustering mutations. Only two of the 168 clusters that contain mutations that are 10bp to 20kb apart also contain an indel within 20kb. So there is no sign that the mechanism(s) that creates the more distant clustering mutations is prone to create indel mutations.

**Mutagenic effect of recombinations.** Recombinations are known to sometimes give rise to large structural variants [30], but it is still uncertain whether they also produce *de novo* SNVs [9]. Using long-range phasing [31] to determine the phase and parental origin of alleles from SNPs typed on Illumina micro-array platforms, we were able to identify the number and location of the crossover events that occurred in the germ cells that gave rise to the offspring [32] in 266 of the 283 families. The resolution to which crossover events can be mapped depends on the distance between flanking heterozygote markers. In our micro-array genotype data, the average bin length that we can map a crossover to is 201kb. We used these data to test for an association between the locations of crossover events and MNM clusters within individuals. As we could not determine the parental origin of all the *de novo* mutations, we considered paternal and maternal events combined. Our results show that three of the mutations in MNM clusters are in the same bin as a crossover event. A permutation test shows that this is not significantly more than would be expected by chance. If we do not restrict the analysis to clustering mutations we observe that 83 of the *de novo* SNVs overlap with a recombination bin in one of the parents. This is 41% more than expected by chance (permutation test p-value = 0.0038). Thus, recombination appears to have a mutagenic effect, but this effect cannot explain the large number of clustering mutations that we observe.

Effect of replication timing. Late replicating regions of the human genome have previously been shown to have a higher SNP density [7] and to harbor more *de novo* mutations than the early replicating regions [11]. We used replication time data from the ENCODE project [33] to investigate the correlation with mutation rate in our data set. This analysis shows that the rate of mutation is significantly increased in late replicating regions both for mutations at CpG-sites and non-CpG sites (see Fig 5A) (combined p-value =  $8.29 \times 10^{-19}$ ). This association is stronger for the clustered mutations than the non-clustered mutations (Fig 5B). In particular, clustered mutations that are 1-10bp apart occur predominantly in late replicated regions of the genome. Contrary to a recent study we do not observe a different effect of replication timing in families with young fathers from that of families with old fathers [11] (See S2 Fig).

# Discussion

Recently there has been much interest in studying the human mutation rate using sequence data from nuclear families [1–5]. With WGS data from 283 Icelandic families, we have the resolution to study relatively rare *de novo* events such as indels and MNM events in more detail than article with smaller sample sizes. In addition to a large number of trios, the present study is also helped by the fact that sequence data are also available from a substantial fraction of the Icelandic population. These population data assist in assessing sequence quality and make it easier to weed out false positives.

Our study confirms previous reports that the rate of mutation is lower than the value of  $1 \times 10^{-9}$  PPPY that has often been used for dating of divergence times within and among humans and related species [16]. Some studies that use other methods have achieved slightly higher rates than the rate we report here [34,35], but studies based on sequencing of parent-offspring trios have consistently yielded estimates close to our estimate [1–4,15]. One implication of this is that some estimates of speciation times between humans and closely related species [36,37] are probably too low and should be adjusted upwards [16]. Our rate estimates were calculated using a probabilistic method that estimates the effective number of positions



**Fig 5.** The effect of replication timing on the mutation rate. a) The effect of replication timing At CpG, non-CpG Strong (C or G) and Weak (A or T) sites. The y-values are mutation rates per position per year (PPPY). The x-values are wavelet-smoothed signal of replication timing calculated by the ENCODE project. Early replicating regions have high signal values and late replicating regions have low values. The bands around the points show the 95% confidence interval for each point. b) The effect of replication timing on clustering and non-clustering mutations. The y-axis shows a combined odds ratio for CpG, non-CpG-Strong and non-CpG-Weak sites calculated using the Cochran-Mantel-Haentzel method. An OR over 1 indicates that we observe most mutations in the latest replicating half of the genome. The error bars are 95% confidence intervals.

doi:10.1371/journal.pgen.1006315.g005

PLOS GENETICS

where a *de novo* mutation could be called [1]. This method adjusts the rate for false negatives caused by insufficient sequence coverage in some regions, but fails to account for mutations that cannot be called regardless of sequence coverage. We do not expect this to have a note-worthy effect on our estimate of the SNV rate, but it could be a problem for indels. Some indels are hard to call because reads containing them fail to be mapped correctly to the reference. It is thus likely that our estimate of the indel mutation rate is too low. Insertions are particularly hard to call based on short-read sequence data and this might explain why we observe a larger ratio of deletions to insertions than expected based on analysis of segregating indels.

We estimate that approximately 3% of human de novo mutations are part of MNM events. This calls into question the accuracy of population genetic models that assume all mutations to be independent events. Incorporating MNMs into methods that use the spatial distribution of mutations to estimate demographic parameters could, for example, make a substantial difference to the accuracy of such models [24]. The prevalence of MNM events also changes how we think about the evolution of functional alleles. If there is a fitness valley between the current amino-acid sequence of a protein and its optimal state, it is important to take the possibility of MNM events into account. Instead of having to take one step backwards and then two steps forward a MNM event makes it possible for the protein sequence to jump directly to the fitness peak in a single step [22]. Previous studies have in fact noticed protein sequence changes that appear to be caused by MNM events. The first report of this phenomenon was the observation that there is a higher rate than expected of switches between TCN and AGY codons at conserved serine residues and it is likely that these codon switches are explained by single synonymous MNM events rather than two independent non-synonymous mutations [19].

The rate of recombination is positively correlated with genetic diversity in the human genome, but it has so far been unknown whether these factors are causatively linked. Our study demonstrates that recombination has a mutagenic effect through increasing the rate of *de novo* SNV mutations. This result is in line with a recent study that reported an increase in genetic diversity around DNA double-strand break (DSB) hotspots [38]. We furthermore report a significantly greater mutation rate in late replicating regions, particularly in the case of MNM events.

Our investigation of MNM events suggests that there might be more than one type of underlying mutational mechanism. Our results corroborate earlier reports that the errorprone polymerase  $\zeta$  is involved in the creation of tandem mutations. In the more distal clustering mutations that are more than 10bp apart we see observe excess of C $\rightarrow$ G mutations, which points to a mutational mechanism involving cytosine deamination followed by error-prone translesional DNA synthesis by REV1 and polymerase  $\zeta$ .

#### Methods

#### Calling de novo variants

The criteria for calling a *de novo* mutation at any given position in trios, where the child is referred to as the proband, were as follows:

- 1. The proband has a genotype likelihood ratio lik (AR)/lik (RR) > 10<sup>10</sup>, where R denotes the reference allele and A the alternative allele.
- 2. For both parents the ratio lik (RR)/lik (AR) > 200.
- 3. None of the 2,636 sequenced individuals (excluding any descendants of proband) have a likelihood ratio: lik (AR)/lik (RR) or lik (AA)/lik (RR) >  $10^4$ . Note that this means that some recurrent mutations could have been filtered out, but it is expected that the number will be small.
- 4. There are at least 15 quality (> = 20) reads for the proband at the mutated site and at least eight reads in each of the parents.
- 5. The number of reads supporting an A allele call should make up at least 30% of the quality sequence reads in the proband.
- 6. The A allele should be observed on both strands in the proband.
- 7. Filter based on site quality metrics calculated by GATK: The ReadPosRankSum value (Z-score approximation of the Wilcoxon rank-sum test on the position of the alternative allele in the reads that support it compared to the position of the reference allele) should be between -6 and 6, the MQRankSum (Z-score approximation of the Wilcoxon rank-sum test of the mapping quality of the read with the alternative allele compared to the reads with reference allele) should be above 6 and the FS value (Fisher's exact test on strand bias) should be below 20.

#### Estimating rates

One possible source of error when analyzing WGS data is that all parts of the genome are not equally well covered by the sequencing. This means that there are parts of a genome where it is hard to correctly call a *de novo* mutation even if it is there. This needs to be taken into account when the denominator in mutation rate estimates is calculated. Instead of making a hard cutoff between callable and non-callable positions in the genomes we use a probabilistic approach to

estimate the number of sites where a *de novo* mutation could be called [1]. The probability of calling site *x* as a *de novo* mutation, given that it is a true de novo mutation in family *f*, we name the *callability* and we denote it by  $C_f^{de novo}(x)$ . The callability is estimated independently for each family conditional on the depth of the three family members at the site using the method described in [1]. The number of callable sites in a given family is then the sum of the callability of all sites in that family. If  $n_f$  denotes the number of de novo mutations found in family *f*, the estimated rate per generation is:

$$rate_{ppp_{G}} = \frac{\sum_{f \in \text{families}} n_{f}}{2 \sum_{f \in \text{families}} \sum_{x \in \text{sites}} C_{f}^{\text{de novo}}(x)}$$

And the estimate of the rate per year is:

$$rate_{pppy} = \frac{\sum_{f \in \text{families}} n_f}{\sum_{f \in \text{families}} ((p_f + m_f) \sum_{x \in \text{sites}} C_f^{\text{de novo}}(x))}$$

Where  $p_f$  and  $m_f$  are the paternal and maternal generation intervals in family *f*.

The callability is estimated separately for SNVs and indels, because indels are generally harder to call than SNVs. The average number of callable sites in a family is 2.45 billion for SNVs and and 2.43 billion for indels. The distribution of the number of callable sites for each family is shown in <u>S3 Fig</u>. When calculating the percentage of *de novo* mutations that we expect to identify, we use the number of non-N bases in hg18 (2.68 billion) and thus ignore the telomeres and centromeres.

Confidence intervals for the rate estimates (and for proportions of mutations) were calculated using the Wilson method [<u>39</u>].

#### Testing the distribution of distances between mutations

To calculate how many mutations are expected to be less than 20kb apart in the same individual, assuming independence, we randomly permuted the family identifier of the mutations 500 times. For each permutation we then counted the number of mutations that was less than 20kb from a mutation in the same individual and found the median of that number over the 500 permutations.

To calculate the distribution of distances between mutations assuming an independent distribution of mutations (Fig 2A), we simulated 500 random mutation sets where each individual had the same number of mutations at "CpG", "non-CpG Strong (C or G)" and "Weak (A or T)" sites as in the original data set. For each of these three types of sites, the positions of mutations were drawn so that the probability that a certain position was mutated was proportional to the mean estimated callability of that position. To ensure that we always have the same number of distances in each simulation, we considered the genome as a one long string of concatenated chromosomes. To calculate the three different kinds of distances used in Fig 2A for a given data set we first sorted the mutations. To find the distance to the nearest mutation in any individual (red line) we then listed the distance between each mutation and the next mutation in the sorted list. To find the distance to the nearest mutation in the same individual (blue line) we for each mutation observed the distance to the next mutation in the list that occured in the same individual. To find the distance to the nearest mutation in a different individual (green line) we observed the distance to the next mutation in the list that was not in the same individual. To produce the values for the x-axis in Fig 2A, we then sorted the distances (for each of the three kinds of distances) for each of the simulated data sets and took the median over the 500 simulated data sets for each rank.

# Testing the effect of replication time on the mutation rate

We downloaded bigwig files of wavelet-smoothed signal of replication timing for five different individuals from the ENCODE project [33,40] (see supplement for list of files) and for each position in the genome we calculated the average value. To avoid spurious correlations based on differences in sequence context, the analysis was stratified into three context groups "CpG", "non-CpG Strong (C or G)", "Weak (A or T)". For each group, we counted the numbers of mutated sites and calculated the sum of the callability for non-mutated sites for each possible replication time value (disregarding sites that did not have replication time data). A p-value was calculated using a logistic regression model with the context group as a covariate. To produce Fig 5A, showing the effect of replication time and the average replication time value was calculated for each of the ten bins. To produce Fig 5B, that shows the effect of replication timing on clustering mutations, we found the median value and split the data into late replicating and early replicating halves. We then used the Cochran-Mantel-Haentzel method to calculate the combined ORs across the three context groups. The same method was used to produce S2 Fig.

# Read-backed phasing of mutations

To investigate whether clustering mutations originated on the same chromosome we used pysam to find all read-pairs that spanned more than one clustering mutation. If at least 90% of the reads spanning a pair of mutations supported a particular phasing we considered that to be the true phasing of the pair.

### Comparing mutation spectra and mutation motifs

We used a Fisher's exact test to compare the distributions of different mutation types shown in Fig 3. A Fisher's exact test was also used when testing for enrichment of a specific kind of mutation motif.

# Testing the effect of recombination on the mutation rate

The recombination analysis was restricted to 266 (out of the 283) families where recombinations had been successfully mapped in both parents. For each family, the number of mutations falling within a recombination bin was counted. Then the family identifier of the recombination lists was permuted, so that the positions of the *de novo* mutations in one family could be compared to the recombination positions from another family. Permutations were performed 5000 times and a p-value was calculated by counting how many times the number of mutations falling within a recombination bin was higher than in the observed data. The expected number of mutations falling within a recombination bin was calculated as the median count in the 5000 permutations.

# **Supporting Information**

**S1 Fig. Length distribution of indels.** (EPS)

**S2 Fig. Effect of replication timing vs. father's age.** (TIFF)

**S3 Fig. Distribution of effective number of callable sites.** (EPS)

S1 Table. The found *de novo* mutations. (XLSX)
S2 Table. Clusters of *de novo* mutations. (XLSX)

# **Author Contributions**

Conceptualization: SB PS AH UT AK DFG KS.

Formal analysis: SB.

Investigation: SB.

Methodology: SB.

Resources: OTM GM.

Supervision: UT AK KS.

Validation: HK AsJ AdJ.

Writing - original draft: SB.

Writing - review & editing: SB PS AH DFG KS.

#### References

- Besenbacher S, Liu S, Izarzugaza JMG, Grove J, Belling K, Bork-Jensen J, et al. Novel variation and de novo mutation rates in population-wide de novo assembled Danish trios. Nat Comms. Nature Publishing Group; 2015; 6: 5969–.
- Kong A, Frigge ML, Masson G, Besenbacher S, Sulem P, Magnusson G, et al. Rate of de novo mutations and the importance of father's age to disease risk. Nature. Nature Publishing Group; 2012; 488: 471–475. doi: 10.1038/nature11396 PMID: 22914163
- Michaelson JJ, Shi Y, Gujral M, Zheng H, Malhotra D, Jin X, et al. Whole-genome sequencing in autism identifies hot spots for de novo germline mutation. Cell. 2012; 151: 1431–1442. doi: <u>10.1016/j.cell.2012</u>. <u>11.019</u> PMID: <u>23260136</u>
- Conrad DF, Keebler JEM, DePristo MA, Lindsay SJ, Zhang Y, Casals F, et al. Variation in genome-wide mutation rates within and between human families. Nature genetics. 2011; 43: 712–714. doi: <u>10.1038/</u> <u>ng.862</u> PMID: <u>21666693</u>
- Jiang Y-H, Yuen RKC, Jin X, Wang M, Chen N, Wu X, et al. Detection of Clinically Relevant Genetic Variants in Autism Spectrum Disorder by Whole-Genome Sequencing. The American Journal of Human Genetics. 2013; 93: 249–263. doi: <u>10.1016/j.ajhg.2013.06.012</u> PMID: <u>23849776</u>
- Ségurel L, Wyman MJ, Przeworski M. Determinants of mutation rate variation in the human germline. Annu Rev Genomics Hum Genet. 2014; 15: 47–70. doi: <u>10.1146/annurev-genom-031714-125740</u> PMID: <u>25000986</u>
- Stamatoyannopoulos JA, Adzhubei I, Thurman RE, Kryukov GV, Mirkin SM, Sunyaev SR. Human mutation rate associated with DNA replication timing. Nature genetics. 2009; 41: 393–395. doi: <u>10.</u> <u>1038/ng.363</u> PMID: <u>19287383</u>
- Hellmann I, Prüfer K, Ji H, Zody MC, Pääbo S, Ptak SE. Why do human diversity levels vary at a megabase scale? Genome Res. 2005; 15: 1222–1231. doi: <u>10.1101/gr.3461105</u> PMID: <u>16140990</u>
- Spencer CCA, Deloukas P, Hunt S, Mullikin J, Myers S, Silverman B, et al. The influence of recombination on human genetic diversity. PLoS Genet. 2006; 2: e148. doi: <u>10.1371/journal.pgen.0020148</u> PMID: <u>17044736</u>
- Webster MT, Hurst LD. Direct and indirect consequences of meiotic recombination: implications for genome evolution. Trends Genet. 2012; 28: 101–109. doi: <u>10.1016/j.tig.2011.11.002</u> PMID: <u>22154475</u>
- Francioli LC, Polak PP, Koren A, Menelaou A, Chun S, Renkens I, et al. Genome-wide patterns and properties of de novo mutations in humans. Nature genetics. 2015; 47: 822–826. doi: <u>10.1038/ng.3292</u> PMID: <u>25985141</u>

- Amos W. Even small SNP clusters are non-randomly distributed: is this evidence of mutational nonindependence? Proc Biol Sci. 2010; 277: 1443–1449. doi: <u>10.1098/rspb.2009.1757</u> PMID: <u>20071383</u>
- Gudbjartsson DF, Helgason H, Gudjonsson SA, Zink F, Oddson A, Gylfason A, et al. Large-scale whole-genome sequencing of the Icelandic population. Nature genetics. 2015.
- Kloosterman WP, Francioli LC, Hormozdiari F, Marschall T, Hehir-Kwa JY, Abdellaoui A, et al. Characteristics of de novo structural changes in the human genome. Genome Res. 2015; 25: 792–801. doi: <u>10.1101/gr.185041.114</u> PMID: <u>25883321</u>
- Wong WSW, Solomon BD, Bodian DL, Kothiyal P, Eley G, Huddleston KC, et al. New observations on maternal age effect on germline de novo mutations. Nat Comms. 2016; 7: 10486.
- Scally A, Durbin R. Revising the human mutation rate: implications for understanding human evolution. Nature Reviews Genetics. 2012; 13: 745–753. doi: 10.1038/nrg3295 PMID: 22965354
- Dong S, Walker MF, Carriero NJ, DiCola M, Willsey AJ, Ye AY, et al. De Novo Insertions and Deletions of Predominantly Paternal Origin Are Associated with Autism Spectrum Disorder. Cell Rep. 2014; 9: 16–23. doi: 10.1016/j.celrep.2014.08.068 PMID: 25284784
- Montgomery SB, Goode DL, Kvikstad E, Albers CA, Zhang ZD, Mu XJ, et al. The origin, evolution, and functional impact of short insertion-deletion variants identified in 179 human genomes. Genome Res. Cold Spring Harbor Laboratory Press; 2013; 23: 749–761. doi: 10.1101/gr.148718.112 PMID: 23478400
- Averof M, Rokas A, Wolfe KH, Sharp PM. Evidence for a high frequency of simultaneous double-nucleotide substitutions. Science. 2000; 287: 1283–1286. PMID: 10678838
- Hodgkinson A, Eyre-Walker A. Human Triallelic Sites: Evidence for a New Mutational Mechanism? Genetics. 2010; 184: 233–241. doi: <u>10.1534/genetics.109.110510</u> PMID: <u>19884308</u>
- 21. Campbell CD, Chong JX, Malig M, Ko A, Dumont BL, Han L, et al. Estimating the human mutation rate using autozygosity in a founder population. Nature genetics. 2012.
- Schrider DR, Hourmozdi JN, Hahn MW. Pervasive multinucleotide mutational events in eukaryotes. Curr Biol. 2011; 21: 1051–1054. doi: <u>10.1016/j.cub.2011.05.013</u> PMID: <u>21636278</u>
- Chen J-M, Cooper DN, Férec C. A new and more accurate estimate of the rate of concurrent tandembase substitution mutations in the human germline: ~0.4% of the single-nucleotide substitution mutation rate. Hum Mutat. 2014; 35: 392–394. doi: 10.1002/humu.22501 PMID: 24375656
- 24. Harris K, Nielsen R. Error-prone polymerase activity causes multinucleotide mutations in humans. Genome Res. 2014.
- Stone JE, Lujan SA, Kunkel TA. DNA polymerase zeta generates clustered mutations during bypass of endogenous DNA lesions in Saccharomyces cerevisiae. Environ Mol Mutagen. 2012; 53: 777–786. doi: 10.1002/em.21728 PMID: 22965922
- Zhong X, Garg P, Stith CM, McElhinny SAN, Kissling GE, Burgers PMJ, et al. The fidelity of DNA synthesis by yeast DNA polymerase zeta alone and with accessory proteins. Nucleic Acids Research. Oxford University Press; 2006; 34: 4731–4742. doi: 10.1093/nar/gkl465 PMID: 16971464
- 27. Rebhandl S, Huemer M, Greil R, Geisberger R. AID/APOBEC deaminases and cancer. Oncoscience. 2015; 2: 320–333. doi: <u>10.18632/oncoscience.155</u> PMID: <u>26097867</u>
- Seplyarskiy VB, Bazykin GA, Soldatov RA. Polymerase ζ Activity Is Linked to Replication Timing in Humans: Evidence from Mutational Signatures. Molecular Biology and Evolution. 2015; 32: 3158–3172. doi: 10.1093/molbev/msv184 PMID: 26376651
- Roberts SA, Lawrence MS, Klimczak LJ, Grimm SA, Fargo D, Stojanov P, et al. An APOBEC cytidine deaminase mutagenesis pattern is widespread in human cancers. Nature genetics. Nature Publishing Group; 2013; 45: 970–976. doi: 10.1038/ng.2702 PMID: 23852170
- Sasaki M, Lange J, Keeney S. Genome destabilization by homologous recombination in the germ line. Nat Rev Mol Cell Biol. 2010; 11: 182–195. doi: <u>10.1038/nrm2849</u> PMID: <u>20164840</u>
- Kong A, Masson G, Frigge ML, Gylfason A, Zusmanovich P, Thorleifsson G, et al. Detection of sharing by descent, long-range phasing and haplotype imputation. Nature genetics. 2008; 40: 1068–1075. doi: 10.1038/ng.216 PMID: 19165921
- Kong A, Thorleifsson G, Frigge ML, Masson G, Gudbjartsson DF, Villemoes R, et al. Common and lowfrequency variants associated with genome-wide recombination rate. Nature genetics. 2014; 46: 11– 16. doi: <u>10.1038/ng.2833</u> PMID: <u>24270358</u>
- ENCODE Project Consortium, Bernstein BE, Birney E, Dunham I, Green ED, Gunter C, et al. An integrated encyclopedia of DNA elements in the human genome. Nature. 2012; 489: 57–74. doi: <u>10.1038/</u> <u>nature11247 PMID: 22955616</u>
- Sun JX, Helgason A, Masson G, Ebenesersdóttir SS, Li H, Mallick S, et al. A direct characterization of human mutation based on microsatellites. Nature genetics. Nature Publishing Group; 2012; 44: 1161– 1165. doi: <u>10.1038/ng.2398</u> PMID: <u>22922873</u>

- 35. Lipson M, Loh P-R, Sankararaman S, Patterson N, Berger B, Reich D. Calibrating the Human Mutation Rate via Ancestral Recombination Density in Diploid Genomes. PLoS Genet. 2015; 11: e1005550. doi: <u>10.1371/journal.pgen.1005550</u> PMID: <u>26562831</u>
- 36. Reich D, Green RE, Kircher M, Krause J, Patterson N, Durand EY, et al. Genetic history of an archaic hominin group from Denisova Cave in Siberia. Nature. 2010; 468: 1053–1060. doi: <u>10.1038/nature09710</u> PMID: <u>21179161</u>
- **37.** Green R, Krause, Briggs A, Maricic. A draft sequence of the Neandertal genome. Science. 2010; 328: 14.
- Pratto F, Brick K, Khil P, Smagulova F, Petukhova GV, Camerini-Otero RD. DNA recombination. Recombination initiation maps of individual human genomes. Science. 2014; 346: 1256442. doi: <u>10.</u> <u>1126/science.1256442</u> PMID: <u>25395542</u>
- **39.** Agresti A, Coull BA. Approximate Is Better than "Exact" for Interval Estimation of Binomial Proportions. The American Statistician. 1998; 52: 119.
- 40. Hansen RS, Thomas S, Sandstrom R, Canfield TK, Thurman RE, Weaver M, et al. Sequencing newly replicated DNA reveals widespread plasticity in human replication timing. Proc Natl Acad Sci USA. 2010; 107: 139–144. doi: <u>10.1073/pnas.0912402107</u> PMID: <u>19966280</u>